Author's response to reviews

Title: Cardiac condition during cooling and rewarming periods of therapeutic hypothermia after cardiopulmonary resuscitation

Authors:

Serdar Demirgan (serdardemirgan@hotmail.com)
Kerem Erkalp (keremerkalp@hotmail.com)
M. Salih Sevdi (salihseydi@yahoo.com)
Meltem Turkay Aydogmus (meltem72_3@hotmail.com)
Numan Kutbay (nkutbay@hotmail.com)
Aydin Firincioglu (aydinfirincioglu@gmail.com)
Ali Ozalp (ali_ozalp3@hotmail.com)
Aysin Alagol (aysinalagol@yahoo.com)

Version: 8
Date: 24 August 2014

Author's response to reviews: see over
Dear Editor;

Our manuscript serial number is “2054454991223597” and its title is “Cardiac condition during cooling and rewarming periods of therapeutic hypothermia after cardiopulmonary resuscitation”… Authors: Serdar Demirgan, Kerem Erkalp, M. Salih Sevdi, Meltem Turkay Aydogmus, Numan Kutbay, Aydin Firincioglu, Ali Ozalp and Aysin Alagol

We thank you for your polite response. We are very happy from your interest to our manuscript.

The manuscript was revised and the comments were added point-by-point responses.

We also hope that this last form of our manuscript is suitable for your readers.

Sincerely yours.

25.08.2014, Istanbul, Turkey

Dr. Kerem Erkalp

On behalf of all the authors.